Author's response to reviews

Title: Effect of changed organisation of nutritional care of Danish medical inpatients

Authors:

Karin O Lassen (k.lassen@dadlnet.dk)
Edvin Grinderslev (edvin@grinderslev.org)
Ruth Nyholm (ruthjohnn@hotmail.dk)

Version: 7 Date: 31 July 2008

Author's response to reviews: see over
Dear BioMed Central Editorial Production Team,

Thank you for the mail. The revised manuscript conforms to the requested revisions in the following way:

- The Reference list contains all authors.
- The figure has now been cropped once again. If this cropping is not acceptable, I will ask you to make this cropping, because I do not know what to do further thanking you in anticipation.

We are looking forward to the publication of the manuscript.

Yours sincerely

Karin O. Lassen
Ruth Nyholm
Edvin Grinderslev